Author’s response to reviews

Title: Determinants of treatment in patients with stage IV renal cell carcinoma

Authors:

Christopher Hollenbeak (csh10@psu.edu)

Eric Schaefer (eschaefe@phs.psu.edu)

Justin Doan (justin.doan@bms.com)

Jay Raman (jraman@pennstatehealth.psu.edu)

Version: 1 Date: 12 Nov 2019

Author’s response to reviews:

Editor Comments

1. Published abstract. We notice that the abstract of this manuscript has been published before: “https://ascopubs.org/doi/abs/10.1200/JCO.2017.35.6_suppl.486”

In your consent for publications statement, please add the statement “The results presented in this paper have not been published previously in whole or part, except in abstract form.” to clearly indicate where the abstract was published (i.e. the name of the conference). The authors must obtain permission from the copyright holder, and this must be stated in the ‘Consent to publish’ statement. Alternatively, you may want to rephrase the Abstract to minimize the overlap.

We have rephrased the abstract so that it differs in form from the Abstract published in the ASCO GU meeting. We have also added the language suggested to the consent for publications statement.

2. Funding. The role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared in this section.

We have clarified the role of the funder in the study design and conduct. In short, the funder gave us latitude to make these decisions and publish as we saw fit.

3. Corresponding author’s email. Please ensure email address of the corresponding author on the manuscript title page match that listed in the editorial submission system.

This has been verified. The correct email address for the corresponding author that is listed in the editorial system is: csh10@psu.edu
5. List of abbreviations. As many abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the Declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

We have added a list of abbreviations, as suggested.

6. Clean copy. Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

We have removed all tracked changes and highlighting. The uploaded copy is clean, as requested.